Author’s response to reviews

Title: Serology based disease status of Pakistani population infected with Hepatitis B virus

Authors:

Muhammad Masroor Alam (ursmasroor@yahoo.com)
Sohail Zahoor Zaidi (zaidis@pak.emro.who.int)
Salman Akber Malik (samalikgau@yahoo.com)
Asif Naeem (drasifnaeem@yahoo.com)
Sahahzad Shaukat (vibgyors@yahoo.com)
Salmaan Sharif (salmaansharif@hotmail.com)
Mehar Angez (meharangez@yahoo.com)
Anis Khan (khvirologist@yahoo.com)
Javed Aslam Butt (drjavedab@yahoo.com)

Version: 6 Date: 27 June 2007

Author’s response to reviews: see over
Cover Letter

1. The word “Address” has been removed.
2. Corresponding author’s symbol and legend has been added.
3. All emails have been added after C. author legend.
4. Figure legend has been corrected.
5. Abstract has been added to the second page after line removal.
6. Main text has been started from page after abstract.
7. Abstract English Quality has been assessed.
8. All underlining has been removed.
9. Table has been moved to after the figure legend.